Author's response to reviews

Title: Screening for hemoglobin Bart's disease among fetuses at risk at mid-pregnancy using the fetal cardiac diameter to biparietal diameter ratio

Authors:

Nopamas Thathan (mamoo_a@hotmail.com)
Kuntharee Traisirisilp (doi_pui@hotmail.com)
Suchaya Luewan (suleuwan@med.cmu.ac.th)
Kasemsri Srisupundit (ksrisupu@mail.med.cmu.ac.th)
Fuanglada Tongprasert (fpunyat@mail.med.cmu.ac.th)
Theera Tongsong (ttongson@mail.med.cmu.ac.th)

Version: 6 Date: 15 June 2014

Author's response to reviews:

Response to the editorial comment:

The text has been changed as suggested (In the abstract: Background, as highlighted in the manuscript.)